Author's response to reviews

Title: Penetrating eyelid injury: a case report and review of literature

Authors:

Ehab Wasfi (wasfiehab@hotmail.com)
B Kendrick (Kendrick@hotmail.com)
T Yasen (Yasen@hotmail.com)
Priya Varma (Varmap4@gmail.com)
Alaa A Abd-Elsayed (alaaawny@hotmail.com)

Version: 2 Date: 9 January 2009

Author's response to reviews: see over
1- The list of authors in the manuscript was written exactly as they are in the submission system.
2- Competing interests section was added
3- Authors’ contributions were added.
4- Authors’ contributions were added.
5- All authors read and approved the final manuscript.
6- Consent section was added
7- Figure legends were moved to a position after the references.
8- Figures were cited as (figure 1) rather than figure (1).

N.B. We give the permission to the journal to make any formatting changes requested.